Author's response to reviews

Title: Determinants of neonatal mortality in Nigeria: Evidence from 2008 Demographic and Health Survey.

Authors:

- Osita K Ezeh (ezehosita@yahoo.com)
- Kingsley E Agho (k.agho@uws.edu.au)
- Michael J Dibley (mdibley@health.usyd.edu.com)
- John Hall (John.Hall@newcastle.edu.au)
- Andrew N Page (a.page@uws.edu.au)

Version: 6
Date: 6 March 2014

Author's response to reviews: see over
Editor-in-Chief
BMC Public Health

Dear Editor-in-Chief

Determinants of neonatal mortality in Nigeria: Evidence from the 2008 Demographic and Health Survey. MS No.:1018342708103101

Ezeh, OK, Agho, KE, Dibley MJ, Hall, J and Page, AN

We enclose the final manuscript of the above article. The article has been edited by English language editor from Edanz group and all changes suggested have been incorporated into the manuscript. The authors have reviewed the final version of the manuscript, approved it for publication and take public responsibility for it. None of the authors have any professional or financial conflicts of interest related to this work.

We look forward to hearing from you in due course.

Yours Sincerely,
Ezeh, Osita Kingsley
ezehosita@yahoo.com